Author’s response to reviews

Title: Intimate Partner Violence among HIV Positive Women in Care - Results from a National Survey, Uganda 2016

Authors:

Steven Ndugwa Kabwama (skabwama@musph.ac.ug)
Justine Bukenya (jbukenya@musph.ac.ug)
Joseph Matovu (jmatovu@musph.ac.ug)
Violet Gwokyalya (vgwokyalya@musph.ac.ug)
Fredrick Makumbi (fmakumbi@musph.ac.ug)
Jolly Beyeza-Kashemya (jbeyeza@yahoo.com)
Shaban Mugerwa (shabanmugerwa@yahoo.com)
John Bwanika (bwanikajb@yahoo.com)
Rhoda Wanyenze (rwanyenze@musph.ac.ug)

Version: 3 Date: 17 Oct 2019

Author’s response to reviews:

BMWH-D-19-00108R2

"Intimate Partner Violence among HIV Positive Women in Care - Results from a National Survey, Uganda 2016"

Steven Ndugwa Kabwama; Justine Bukenya; Joseph KB Matovu; Violet Gwokyalya; Fredrick Makumbi; Jolly Beyeza-Kashemya; Shaban Mugerwa; John Baptist Bwanika; Rhoda K. Wanyenze

BMC Womens Health
Dear Dr. Amanpreet Athwal,

On my own behalf and on behalf of all the other co-authors, I would like to thank you for reviewing our manuscript, "Intimate Partner Violence among HIV Positive Women in Care - Results from a National Survey, Uganda 2016" (BMWH-D-19-00108R2), and giving us the opportunity to improve it. Below this letter are point by point responses to the comments raised by the associate editor:

Assistant Editor Comments:

1. Title

The title is different in the main manuscript and submission system. Please amend one so they match.

Response: The two titles have been revised to match.

2. Title page

We notice a mismatch of the author Rhoda K. Wanyenze's email between the title page and the submission system. Please amend this.

Response: The two emails have been revised to match.

3. Authors' Contributions

We note you have included the initials [JA] in the Authors' Contributions section which does not match the authors on the author list. If the order of authors has changed since original submission of the manuscript. In line with COPE guidelines, BioMed Central requires written confirmation from all authors that they agree with any proposed changes in authorship of submitted manuscripts or published articles.

Response: The order of authors has not changed. The initials [JA] were a typo and have been removed. We thank the editor for identifying this error.